Author’s response to reviews

Title: Admission C - reactive protein after Acute Ischemic Stroke is associated with Stroke Severity and Mortality: The Bergen Stroke Study.

Authors:

Titto T Idicula (idic@helse-bergen.no)
Jan Brogger (jan.brogger@helse-bergen.no)
Halvor Naess (haln@helse-bergen.no)
Ulrike Waje-Andreassen (uwan@helse-bergen.no)
Lars Thomassen (ltho@helse-bergen.no)

Version: 6 Date: 16 April 2009

Author’s response to reviews:

Dear Editor,

Thanks for your comments to improve the quality of the manuscript. We have made following changes to the manuscript.

1. The title and figure number removed from the figure.
2. Tables are included in the manuscript and additional files are deleted.
3. Used sentence case in the title.
4. ‘Author contribution’ is formatted as a paragraph rather than as a list
5. Removed any blank pages

Please check if you are satisfied with the changes. If in case you need any further changes, please let us know.

Your’s faithfully,

Dr. Titto Idicula (for authors)